Author's response to reviews

Title: Feasibility and Acceptability of a Novel, Computerized Screening and Brief Intervention (SBI) for Alcohol and Sweetened Beverage use in Pregnancy

Authors:

Madhabika B Nayak (mnayak@arg.org)
Rachael A Korcha (rkorcha@arg.org)
Lee A Kaskutas (lkaskutas@arg.org)
Lyndsay A Avalos (Lyndsay.A.Avalos@kp.org)

Version: 5 Date: 16 October 2014

Author's response to reviews: see over
October 16, 2014

Dr Peter O'Donovan
Editor BMC Pregnancy and Child Birth

Dear Dr. O'Donovan:

I write on behalf of my co-authors Drs. Korcha, Kaskutas, and Avalos. We respond here to the email received on October 14, 2014 regarding our manuscript (MS: 1988596258125210) “Acceptability and Feasibility of a novel, computerized screening and brief intervention for alcohol and sweetened beverage use in pregnancy”. We are pleased that our manuscript is viewed as suitable for publication.

We have revised and formatted our manuscript as per each editorial comment listed in the email referred to above from the BioMed Central Editorial office as follows:

1. We have included a Competing interests section between the Conclusions and Authors’ contributions.
2. We include an Authors’ contributions section before the Acknowledgements and Reference list.
3. We have included an Acknowledgements section between the Authors contributions section and Reference list. This section lists the sources of funding for our study and acknowledges the contribution of staff and clients at the clinic where we collected data, without which study completion would not have been possible.
4. We have checked to make sure that our revised manuscript conforms to the journal style specified at http://www.biomedcentral.com/info/ifora/medicine_journals and have formatted files as needed.

Please note that in the process of ensuring that all tables are presented in portrait orientation as specified in the authors’ checklist for manuscript formatting, Tables 2 and 4 had to be re-formatted and differ in appearance from the Tables submitted with prior revisions which were approved by reviewers. To allow for decisions regarding print versions for publications to consider using the wider format tables, we have uploaded the tables in landscape format as originally submitted as a separate additional file.

Thanking you,

Madhabika B. Nayak, Ph.D.
Associate Scientist